Author’s response to reviews

Title: A Logistic Regression Analysis of Risk Factors in ME/CFS Pathogenesis

Authors:

Eliana Mattos Lacerda (Eliana.Lacerda@lshtm.ac.uk)
Keith Geraghty (KeithGeraghty2@gmail.com)
Caroline Kingdon (Caroline.Kingdon@lshtm.ac.uk)
Luis Nacul (Luis.Nacul@lshtm.ac.uk)

Version: 3 Date: 09 Sep 2019

Author’s response to reviews:

Dear Editor,

Thank you for your message and for considering our manuscript. Please find the final version of our manuscript with the requested changes, as requested below:

1. Please remove all information related to funding from the Acknowledgments subsection.
   Answer: the information is now removed and the acknowledgments are adequately updated.

2. Please provide more justification for the contributions of authors.
   Answer: all the authors' contributions to the manuscript are now described.

3. In your Funding section, please also state the role of the funding body in the design of the study; collection, analysis, and interpretation of data; and in writing the manuscript.
   Answer: the funding agencies did not influence the manuscript at any stage, and this information is explicit in the paper now.

4. In the Ethics approval and consent to participate subsection of the Declarations please provide the full unabbreviated name of the ethics committee which approved your study protocol.
   Answer: the abbreviation is now spelled out.

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.
   Answer: files uploaded as requested.

Sincerely,

Eliana
Dr. Eliana Lacerda
LSHTM